Contents lists available at ScienceDirect

# Heliyon

journal homepage: www.heliyon.com

## Research article

# Hybrid AC/DC microgrid test system simulation: grid-connected mode

Leony Ortiz<sup>a,\*</sup>, Rogelio Orizondo<sup>a,\*\*</sup>, Alexander Águila<sup>a,\*\*\*</sup>, Jorge W. González<sup>b</sup>, Gabriel J. López<sup>b</sup>, Idi Isaac<sup>b</sup>

<sup>a</sup> Carrera de Ingeniería Eléctrica, Grupo de Investigación GIREI, Universidad Politécnica Salesiana, Quito, Ecuador
<sup>b</sup> Carrera de Ingeniería Eléctrica, Grupo de Investigación TyD, Universidad Pontificia Bolivariana, Medellín, Colombia

#### ARTICLE INFO

Keywords: Electrical engineering System diagnostics Power system operation Power converter Smart grid technology Distributed resources Microgrid benchmark Hybrid energy systems Power flow

### ABSTRACT

In this paper, a Microgrid (MG) test model based on the 14-busbar IEEE distribution system is proposed. This model can constitute an important research tool for the analysis of electrical grids in its transition to Smart Grids (SG). The benchmark is used as a base case for power flow analysis and quality variables related with SG and holds distributed resources. The proposed MG consists of DC and AC buses with different types of loads and distributed generation at two voltage levels. A complete model of this MG has been simulated using the MATLAB/ Simulink environmental simulation platform. The proposed electrical system will provide a base case for other studies such as: reactive power compensation, stability and inertia analysis, reliability, demand response studies, hierarchical control, fault tolerant control, optimization and energy storage strategies.

## 1. Introduction

Renewable energy systems (RES) propose a new technology that is cleaner and capable of supplying the growing electricity demands of interconnected and isolated communities. In recent years, MGs have become a great attraction for the scientific community as well as a promising solution for future traditional energy systems. MGs are seen as a possible technology for the integration of variable renewable energy systems in the traditional grid.

Currently, with the evolution of new digital technologies, such as micro-processed systems and advances in power electronics, many applications have been implemented in SG, specifically in the development of controllers and electronic energy converters. In recent decades, researchers have made significant contributions which have had a high impact in these areas, mainly aimed at data acquisition, automation, and control of MGs [1, 2, 3]. MGs not only integrate the distributed generation to the Main Grid in a reliable and clean fashion, but also provide high reliability in its capacity to operate in the face of natural phenomena and active Distribution Grids, which in turn results in less energy losses in transmission and distribution and less construction and investment time [3, 4, 5, 6]. Research developed in [7, 8, 9, 10, 11, 12, 13, 14, 15] show actual implemented MGs. Some of the examples can be seen in CERTS in

the US, NEDO in Japan, and a vast majority of MGs in Europe.

A MG can be defined as a low-voltage distribution power system to which small modular generations systems, such as renewable energy sources, other distributed generators, as well as intermediate storage units are connected and can fulfill the load demand. This particular power system can be treated by the utility grid as a controllable load or generator [16]. Although MG configurations can be exclusively DC, AC, or a hybrid of the two technologies, some investigations are particularly focused on AC MG. This is due to its ability to operate together with the Main Grid [4, 17, 18, 19, 20, 21, 22]. It is known that each have several advantages which in turn leaves a HMG benchmark a turning point in this investigation [18, 23].

All MG technologies must face the dynamics and steady state characteristics of the distribution generators (DG), the unbalance and nonlinearity of loads and the proper dynamics of energy storage systems (ESS) [17]. HMGs must also face the problem of an accidental or a programmed disconnection from the Main Grid. An HMG benchmark must be subjected to two typical scenarios, as in any other power system Distribution Grid: minimum and maximum demand situations. HMGs must manage abnormal operations of its electrical infrastructure. A set of three-level control structures is a well-known strategy to control MG parameters [17, 30]. Specialized literature has reported implemented

E-mail addresses: lortizm@ups.edu.ec (L. Ortiz), rorizondo@ups.edu.ec (R. Orizondo), aaguila@ups.edu.ec (A. Águila).

https://doi.org/10.1016/j.heliyon.2019.e02862

Received 27 August 2019; Received in revised form 30 September 2019; Accepted 13 November 2019

2405-8440/© 2019 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).





<sup>\*</sup> Corresponding author.

<sup>\*\*</sup> Corresponding author.

<sup>\*\*\*</sup> Corresponding author.

and experimental MGs in Europe, Africa, Asia and America [31, 32, 33, 34, 35, 36]. In fact, there is also a discussion about its acceptance [37]. Power electronics are included in MG configurations due to the nature of most renewable generation technologies. It is necessary to control the injected power delivered to the MG [17, 23]. If appropriate closed-loop control strategies are implemented, its power quality problems may be overcome [24, 25, 26]. The performance of the MG configurations and different technologies may be improved by the use of parallel inverters [27].

In this study, a detailed model of a Hybrid Microgrid (HMG) benchmark has been simulated. This model is based on the original IEEE-14distribution-bus model. The proposed benchmark does not offer any wind energy resource since the effort is focused on the ability of the MG to operate one of two renewables energies. This is usually the case, as one of the two energy resources is more available than the other. The main objective of providing this benchmark is to set up a complete and detailed model for further studies: reliability and resilience, optimization, fault diagnosis, system identification, and fault tolerant control. There are many control techniques for the operation of controlled rectifiers or inverters, as interfaces to renewable energies. In this paper, some converters have been settled to operate in an open-loop control strategy and others operate in a closed-loop control strategy. Additionally, these rectifiers or inverters use pulse width modulation techniques of different carrier frequencies [17, 28, 29].

The MG is provided with typical balanced and unbalanced loads, linear and non-linear loads, energy storage systems, as well as distribution transformers and line impedances. This paper also offers a thorough description of the model with all the necessary data to tackle the aforementioned studies. Both demand scenarios are analyzed and a discussion about raised and potential problems and their possible solutions are also discussed. This study furnishes all power flow results for both scenarios. Next, the results show the conflicts that these variables may present with the bidirectionality of power flows and with the insertion of predominantly solar photovoltaic distributed generation. Lastly, the present paper is a starting point for the analysis of many current issues that will be described in future research.

The paper is distributed as follows: Section 2 explains the main electrical characteristics of MGs and its architecture. Section 3 describes every component of the proposed MG benchmark. In its subsections, details are given for the photovoltaic subsystems (3.1) and the battery energy storage subsystems (3.2). There is a complete description of the AC/DC voltage source converters (VSC) outside the DC bus (3.3 and 3.8) as well as their coupling transformers (3.4). Subsections 3.5, 3.6 and 3.7 are devoted to outline the electric characteristics of primary and secondary lines in addition to both linear and non-linear loads. Subsection 3.8 sketches the DC bus and all its parts, including their dual DC to AC converters and both a boost converter and a buck-boost converter. Section 4 shows load-flow solutions obtained from the simulation in MAT-LAB/Simulink. The results have been tabulated for the two opposite situations: a maximum demand scenario and a minimum demand scenario. Section 5 analyses both scenarios and future research areas are shown in Section 6. Finally, the conclusions can be found in Section 7.

#### 2. Electric microgrids

MGs, even on a smaller scale, represent one of the most interesting solutions for researchers. MGs improve power flow in Distribution Grids and reduce power losses in transmission lines through DG interconnection, renewable energy sources (RES), battery energy storage systems (BESS) and loads [4, 12]. In the literature, several reviews direct to different control strategies [38, 39, 40, 41, 42, 43, 44], such as: test beds [8, 12], optimization techniques, and available software tools [45, 46], among others. Although MGs are considered on a small scale, their technical complexity on modelling and simulation is higher compared to the conventional energy system. Therefore, models that allow dynamic analysis are a key point to ensure that future MGs work in a stable way.

## 2.1. Microgrid architectures

As shown in Fig. 1, MGs are constituted by a series of systems and subsystems such as: distributed generation, energy storage and different types of loads. The MGs can operate in parallel to the Main Grid, with no power exchange; in isolated mode, with autonomous power supplies; and in an interconnected mode, where it assumes the Main Grid set points [1, 3, 4, 5, 11, 20, 38, 41, 47, 48, 49].

On the other hand, MGs can also be classified by attending to the following types:

- Types of voltage: DC; AC and hybrid.
- Distribution configuration: single-phase; three-phase; three-phase + neutral.
- Voltages: low (LV) and medium (MV).
- Structures: radial and ring.

In the literature, three definitions are listed: Microgrid, Nanogrids, and Picogrids. As described in [7] and shown in Fig. 2, AC/DC MGs are usually implemented in series, switched, parallel or in combination. For the MG series configuration (Fig. 2a), there is a DC bus where all the generation systems and loads are connected through their respective converters.

The parallel configuration (Fig. 2b), has an AC bus where the generation systems and the loads are directly connected. The DC devices are connected through their own inverters, or through a DC bus coupled to the AC bus through an inverter or bidirectional VSC. Finally, in the switched configuration (Fig. 2c), the load can be supplied by a DG source or by the Distribution Grid (never in both cases at the same time), and both the DC MG and the AC MG are connected by two inverters [7, 47].

The configuration for MGs are mostly used in AC. However, DC MGs have gained much interest in recent times due to the advantages offered, such as a lack of reactive power and harmonics. Also, there is no need to synchronize the DC generation, for it presents few power losses, and there are no changes in the DC bus after a blackout. However, despite of all these advantages, disadvantages are present as well, and mainly, is the absence of crossing points by zero and the fact that the protection systems are more complex for high voltage levels [7].

As shown in Fig. 3, the combination between both configurations gives rise to the concept of AC/DC HMGs while proposing an improved approach that combines the main advantages of the AC and DC MGs [4, 20]. Future trends show a greater requirement and effort in research of characteristics such as: scalability, identification and modelling, design, and control structures that allow an integration of HMGs to the Main Grid [20, 21].

Some papers focus mainly on AC MGs and DC MG architectures. HMGs may represent an interesting solution in which the possible advantages of AC and DC configurations could be combined. The AC MGs can be interconnected (coupled) or isolated (decoupled). Two approaches have been identified in decoupled MGs: fully isolated topology or partly isolated topology.

On the other hand, in decoupled AC three topologies are defined: the completely isolated two-stage configuration, the two partially isolated stages and the configuration of three partially isolated stages. The configuration to be defined for the AC/DC HMG greatly depends on the application and the environment in which it is integrated [4, 5, 20, 21]. The main characteristic of the interconnected configuration is that the AC MG is directly connected to the Main Grid by means of an AC transformer, fixing the MG to the grid [20]. The transformer is located at the point of connection with the electrical grid providing galvanic isolation to the entire MG and reducing the voltage level for the low voltage AC grid. The interconnected HMGs are also capable of managing bidirectional power flow (generation/consumption) using bidirectional AC/DC VSC for the case of BESS storage systems. It is important to emphasize that HMGs are easily scalable and their implementation can be done in several levels and configurations. Additionally, HMGs can be integrated



Fig. 1. Electrical MG.



Fig. 2. AC/DC MG configurations.

into a medium-voltage (MV) Distribution Grid or in a low-voltage (LV) residential level [20].

When designing an AC/DC HMG, it is necessary to consider some requirements that must be met, such as reliability, controllability, observability, economy and flexibility. The following design principles from literature are taken into account [5].

- The principle of partition.
- The principle of hierarchy.
- Full use of the resources.

- Maximizing the use of resources.
- Energy complementary.
- Power quality assurance principles.
- Storage allocation.
- Reactive power compensation.
- Configuration: Radial and ring.

## 2.2. Microgrids models and benchmark. A review

The MG models presented in the literature show different



Fig. 3. AC/DC HMG.

characteristics depending on their configuration, type of topology, and components. All of these must deal with dynamics, low energy storage capacity of BESS, high number and diversity of generation, electronic power converters and a high degree of non-linear phenomena [1, 10, 47, 50]. Others seek to model each distributed generation source by reducing the order of the model to a linear and time invariant (LTI) system with a time constant and a gain factor, while neglecting the dynamics of the grid [10, 47].

There are also other approaches with inverter-based DG, for example, in [51] the complete dynamic model of the entire grid was considered in place of the inverter, dividing the MG system into three subsystems: inverter, grid and loads. For the inverter model, the dynamics of the controller, the output filter and the coupling inductor are incorporated. In another paper the authors also develop a model in the LTI linear state space and perform an eigenvalue analysis to study the dynamic behavior of the MG [52]. In [7, 8, 9, 10, 11, 12, 13, 14, 15] real MGs have been implemented all over the world. Also, the authors show a summary of some existing test systems where they include simulated grids and present a comparative summary of them.

### 3. Proposed model

This section presents a detailed description of the proposed threephase model. It was simulated using the corresponding Simscape library within the MATLAB/Simulink R2017b environment. The aim is to provide a tool for the scientific community and offer a better understanding of the MG dynamics and each of its components, as well as their overall performances under different operating conditions.

Fig. 4 shows a one-line diagram for the proposed benchmark. In this figure, the MG is coupled to the PCC point to a 69-kV electrical sub-transmission system. This utility grid has a Thevenin equivalent of 100 MVA with an X/R ratio of 10. There are two voltage distribution levels: a primary 13,8-kV voltage level, depicted in blue color and a secondary 0,22-kV voltage level depicted in green color. Three sub MGs are shown: AC MG 1 is an area connected at the 0,22-kV level through lines 4, 5 and 6 to the AC MG 2. Switches S1, S2 and S3 are considered connected in this paper. It operates with a diesel generator and provides energy to 4 loads. AC MG 2 is another area, which operates with the PV Array 2 and the BESS 2. Both AC MGs operate at a frequency of 60 Hz. The third area is a

DC busbar comprising the BESS 1 and the PV Array 1. The BES system 1 is connected through a boost-buck bidirectional converter, while the PV Array 2 is connected to the DC link through a boost converter. The DC busbar is linked to the AC MG 2 through two parallel bidirectional converters tagged 1.1 and 1.2 which can operate as rectifiers or inverters to exchange active and reactive power through two transformers named TDC-1 and TDC-2. In the one-line diagram, an equivalent bidirectional converter has been placed instead of the two original converters. So is the case of transformer TDC 1–2 which takes the place of both transformers TDC-1 and TDC-2. These simplifications to the one-line diagram are made for the sake of simplicity. Finally, this AC/DC HMG also includes a diesel generator and a Main Grid equivalent system with linear and nonlinear loads.

## 3.1. Solar-Photovoltaic subsystems (PV)

Solar-Photovoltaic technology is one of the distributed renewable energy resources in the MG. Table 1 shows main parameters, which are used by the MATLAB/Simulink environment to characterize performance of the solar cell panels. Both arrays operate with total irradiance on the solar cells of G = 1000 W/m<sup>2</sup> and a temperature of 25 °C. Array 1 has 42 modules which develops a nominal power of 10,5 kW whereas array 2 has 1750 modules, which develops a nominal power of 725 kW for the DC Bus. Both DC-type sources of energy have need of power converter interfaces. Array 1 operates with a DC-DC voltage-controlled boost converter. It has a 6000  $\mu$ F link capacitor operating at 300 VDC. A PWM closed-loop control strategy utilizes a frequency of 5 kHz. Array 2 operates with an open-loop sinusoidal PWM inverter. Its m index is set to 1.0 and steps down from 480 VDC to 250 VAC.

#### 3.2. Battery energy storage systems (BESS)

BESS #1 operates in the DC bus. It consists of 1 lithium-ion battery unit of 120 VDC nominal voltage with a rated capacity of 800 (Ah). BESS #2 operates near LV Bus 1. It consists of 3 nickel-metal-hydride (Ni–MH) battery units of 650 VDC nominal voltage each one with a rated capacity of 1,5 (Ah). The parallel-connected batteries are connected to an interfaced inverter in a cascaded topology which in turn steps up from 650 VDC to 900 VAC. All these battery-types are devices available in the



Fig. 4. Proposed MG.

# Table 1 PV arrays for the MG system

variage for the me system								
ID	Current at maximum power point I <sub>mpp</sub> (A)	Maximum Power (W)	Open circuit voltage V <sub>oc</sub> (V)	Voltage at maximum power point V <sub>mp</sub> (V)	Short- circuit current I <sub>sc</sub> (A)			
Array 1	8,63	250	37,4	30,7	8,63			
Array 2	5,69	414,8	85,3	72,9	6,09			

Simscape library within the MATLAB/Simulink environment.

The Simscape library integrates Eqs. (1), (2), (3), and (4) which govern the discharge-charge process of the nickel-metal-hydride batteries [53]:

$$E_{disch}^{\text{Ni-MH}} = Eo - k \frac{Q}{Q - it} i^* - k \frac{Q}{Q - it} it + Exp(t)$$
<sup>(1)</sup>

$$E_{ch}^{\text{Ni-MH}} = Eo - k \frac{Q}{|it| - 0.1Q} i^* - k \frac{Q}{Q - it} it + Exp(t)$$
(2)

The discharge condition implies  $i^* > 0$  while the charge condition implies  $i^* < 0$ . On the other hand, the equations that govern the discharge-charge process of the lithium-ion batteries are:

$$E_{disch}^{\text{Li-Ion}} = Eo - k \frac{Q}{Q - it} i^* - k \frac{Q}{Q - it} it + AExp(-Bit)$$
(3)

$$E_{ch}^{\text{Li-Ion}} = Eo - k \frac{Q}{|it| - 0.1Q} i^* - k \frac{Q}{Q - it} it + AExp(-Bit)$$
(4)

The discharge-charge conditions are the same as those of the Ni–MH battery type. In the equations, the variables and parameters are:

*E*<sub>o</sub> is the battery constant voltage;

K is the polarization constant (Ah-1);

Exp (t) is the exponential zone dynamics, in V;

Q is maximum battery capacity, in Ah;

it is extracted capacity or the actual battery charge, in Ah;

i\* is the filtered low frequency current dynamics, in A;

A is exponential voltage, in V;

B is exponential capacity, in  $(Ah)^{-1}$ .

Fig. 5 shows the typical constant current/discharging voltage characteristic curve for the lithium-ion battery. Fig. 6 shows the corresponding curve for the nickel-metal-hydride battery. For both curves, three regions are clearly distinguishable: the exponential region in yellow, the nominal or rated region colored in gray and the final discharge behavior in blue [54].

### 3.3. DC-AC voltage source converters

The solar PV array #2 is interconnected to the AC 13,8 kV system through an inverter which steps down from 480 VDC to 250 VAC. Transformer TPV is used to connect the AC inverter side to the MG primary distribution system. The converter is modelled using a 3-level PWM-controlled IGBT bridge. The PWM carrier frequency is 1980 Hz operating at a closed-loop control strategy for both scenarios.

The BESS #2 is also interconnected to the AC 0,22 kV system with an inverter which steps up from 650 VDC to 900 VAC. Transformer TB is used to connect the AC inverter side to the MG secondary distribution system. The inverter is modelled using a multilevel topology determined to be the most efficient and reliable, considering power losses, total harmonic distortion and efficiency. In a recent paper, five different topologies were contrasted and the CMLI (Cascade h bridge Multi Level Inverter) was chosen [55].

A phase shifted SPWM technique is used in an open loop strategy with a carrier frequency of 2500 Hz. In the maximum demand scenario, the inverter generates to the AC inverter side using an overmodulation index of 1,2 whereas in the minimum demand scenario, the inverter can charge the BESS #2 unit from the AC inverter side using a modulation index of 0,8.

## 3.4. Transformers proposed for the microgrid system

There are 7 power transformers in the main case study. A 69 kV/13,8 kV substation transformer, named T3, has a series equivalent impedance of 1,5% on its MVA base and has a delta-wye grounded configuration. It connects the distribution MG to the 69 kV sub-transmission system. The upstream utility grid has a Thevenin equivalent of 100 MVA with an X/R ratio of 10.

Transformers T1 and T2 are used to step down from the 13,8-kV industrial or commercial usage to the 0,22-kV residential usage. On the other hand, transformers TB, TG, TPV and TDC1-2 are used with the energy storage systems and the renewables to interface with the inverters from the MG. A step-up transformer, named TG, connects the distribution diesel generator to the grid. A complete detail of transformers and their ratings are shown in Table 2.

#### 3.5. Line data

Distribution lines for the benchmark MG are typical in AC medium and low voltage distribution levels. For the 13,8-kV primary grid, a 1/ 0 Cu bare conductor was chosen. The unit resistance of the line is 0,394  $\Omega$ /km, the unit reactance of the line is 0,1168  $\Omega$ /km and the unit impedance of the line is 0,411  $\Omega$ /km. For the voltage level 0,22-kV secondary grid, a 4/0 Cu TW cable was chosen. The unit resistance of the line is 0,198  $\Omega$ /km, the unit reactance of the line is 0,1089  $\Omega$ /km and the impedance unit of the line is 0,227  $\Omega$ /km Table 3 gives more details:

#### 3.6. Loads

Table 4 shows the data loads connected to the system, its power factors and unbalanced voltage percentages. Balance and unbalance linear loads are modelled as constant impedances. Loads connected to buses 2 and 3 contain single-phase components that deteriorate the symmetry of voltages and currents in this system. The unbalanced percentage of the loads can be calculated as the percentage value between the maximum deviation of the current of one of the phases with respect to the average load current, as shown in (5) and (6).

$$UL\% = \frac{max\left\{\left|I_a - I_{average}\right| \; ; \; \left|I_b - I_{average}\right| \; ; \; \left|I_c - I_{average}\right|\right\}}{I_{average}} * 100$$
(5)

$$I_{average} = \frac{I_a + I_b + I_c}{3} \tag{6}$$

where:

 $I_a$ ,  $I_b$ ,  $I_c$  are the line rms currents in phases a, b and c.

## 3.7. Non-linear load

A three-phase SCR bridge rectifier operates as a non-linear load at MV Bus9. In the maximum demand condition, an open-loop PWM strategy, 4080 Hz carrier frequency with an  $m_{index}$  of 0,8 defines the operation of the converter with a DC load of 1084  $\Omega$ . In the minimum demand condition, with unaltered PWM parameters, a DC load of 3615  $\Omega$  is placed instead. The rest of parameters are shown in Table 5.

## 3.8. DC bus (PV, BES systems, DC/DC and DC/AC VSC)

As shown in Fig. 7 and explained below, the DC MG bus operates in two opposite situations: a scenario of maximum and minimum demand. In the first operating mode (maximum demand), the PV-1 system can operate to track the maximum power point (maximum power point tracking control - MPPT), and therefore, the AC MG functions as an energy storage of a certain capacity. In the second mode (minimum demand), there is no power generation in the PV-1 system, so the link to the AC MG #2 through the bidirectional AC/DC VSC control the power flow from the AC MG #2 to the DC MG, allowing the storage of energy in the BESS #1 and supply the load in the DC system.

The DC bus has a 2-kW load in the maximum demand scenario. It was modelled as a linear resistance. In future investigations, it can be replaced with a charging station model for electric vehicles (plug-in hybrid electric vehicle, PHEV or electric vehicle, EV). In this perspective, with a SG deployment, the vehicle to grid (V2G) can also become an



Fig. 5. Nominal current discharge curves for the lithium ion battery (BESS #1) at 0.43C (347 A).



Fig. 6. Nominal current discharge curves for one nickel-metal-hydride (Ni-MH) battery (BESS #2) at 0.2C (0.3A).

Table 2

Transformer ratings for the MG system.

Transformer	Nominal Power (kVA)	Voltage Ratio (HV/ LV)	R <sub>cc</sub> (pu)	X <sub>cc</sub> (pu)
T1	1500	Y 13800/220 Y	0,03	0,03
T2	1500	Y 13800/220 Y	0,03	0,03
T3	4000	Yg 69000/13800 D1	0,015	0,015
TB	55	D1 900/220 Y	0,003	0,06
TG	3500	Yg 13800/2400 D1	0,015	0,015
TPV	1000	Yg 13800/250 D1	0,0012	0,03
TDC1-2	15	Y 220/150 Y	0,03	0,06

Table 3

Line data for the MG system.

Line	Sending end	Receiving End	R (ohm)	X (ohm)	Distance (km)
1	LV1	LV2	0,0297	0,016335	0,15
2	LV1	LV5	0,0396	0,02178	0,2
3	LV2	LV5	0,0297	0,016335	0,15
4	LV2	LV4	0,0792	0,04356	0,4
5	LV4	LV5	0,0792	0,04356	0,4
6	LV2	LV3	0,0792	0,04356	0,4
7	LV3	LV4	0,0198	0,01089	0,1
8	MV7	MV9	0,788	0,2336	2,0
9	MV6	MV11	2,364	0,7008	6,0
10	MV6	MV12	2,364	0,7008	6,0
11	MV6	MV13	1,182	0,3504	3,0
12	MV10	MV11	2,364	0,7008	6,0
13	MV13	MV14	1,182	0,3504	3,0
14	MV9	MV14	0,788	0,2336	2,0

Table 4
---------

Load data for the MG System.

Bus ID	Load Name	Load Type	Max. load (kVA)	Min. load (kVA)	PF	Unbalance load (%)
LV2	Load 2	Unbalanced load	40	12	0,9	13
LV3	Load 3	Unbalanced load	30	9	0,85	12,6
LV4	Load 4	Linear load	50	15	0,9	0
MV9	Load 9	Non-linear load	320	96	1	0
MV10	Load 10	Linear load	800	240	0,8	0
MV11	Load 11	Linear load	400	120	0,8	0
MV12	Load 12	Linear load	800	240	0,8	0
MV14	Load 14	Linear load	1600	480	0,8	0
DC	DC load	-	2	0,6	1	-

Tabl	e 5		
Data	for the	non-line	ar Load

PWM	Capacitor	V <sub>n</sub>	R [P <sub>max</sub> ]	R [P <sub>min</sub> ]	m <sub>index</sub>	phi
Freq. (Hz)	(µF)	(V <sub>dc</sub> )	(Ω)	(Ω)	-	(°)
4080	10	18500	1084	3615	0,8	-30

agent that operates as an energy provider [19, 37, 56].

The DC bus converters were modelled as a bidirectional DC/DC converter for the BESS #1 and a DC/DC amplifier controller for the PV #1. Details of this DC MG are given in Table 6. This arrangement was inspired by the model proposed in [57].

Output voltages from the distributed generation sources, as PV systems, are typically intermittent by nature. These systems are totally dependent on random phenomena such as changes in irradiation and temperature that depend on weather conditions during the day. Therefore, it is essential to connect both to the DC bus through a controlled DC-DC converter in order to regulate the output voltage or to follow the maximum power point (MPPT). A topology was chosen for the DC/DC boost converter that is intended to increase the input DC voltage (V<sub>i</sub>) in a range of 100 V–130 V up to a DC output voltage (V<sub>o</sub>) about 350 V, which is the DC bus voltage level, as recommended by the authors in [57]. The switching frequency *f* is set to 5000 Hz in order to avoid two typical problems: for one thing, the audible noises and for another to avoid the high-frequency parasitic elements. To maintain the controller stability, both parameters are selected above the critical values already discussed in [57, 58]. These critical values are given by Eqs. (7) and (8):

$$L_c = \frac{D\left(1-D\right)}{2f} R \tag{7}$$

$$C_c = \frac{D}{2fR}$$
(8)

where:

(

 $L_c$ , is the inductance critical value;  $C_c$ , is the capacitance critical value;

 $\boldsymbol{D}$  , is the large-signal duty cycle;

f , is the switching frequency;

*R* , is the load resistance.

In this case, capacitance is set to 1200  $\mu F$  and the inductance is set to 9 mH, considering a DC resistance of 45  $\Omega$  which corresponds to a power consumption of 2 kW at nominal voltage.

In order to control the loading and unloading processes of the storage device BESS #1, another bidirectional DC-DC controller interfaces two voltages, the DC bus voltage and the battery system voltage. Two PI controllers are intended to achieve the reference current signal in case of loading and unloading. The controller follows a current reference signal to charge and discharge the BESS #1, which is connected to the DC MG.



Fig. 7. DC MG.

Table 6

DC converter parameters and specifications.

DC Bus Converter	PWM Frequency (Hz)	Capacitor (uF)	$V_n (V_{dc})$
Bidirectional Boost-Buck	5000	1200	300
Boost	5000	1200	300

The aim is to target low current ripples in order to achieve greater efficiency and increase the useful life of the battery system. It would also be useful to know that when the MG goes into isolated mode, the control strategy regulates the DC bus voltage [57].

Eqs. (9), (10), (11), and (12) are considered in order to select the filter values. BESS batteries are simulated according to characteristics presented in Table 7, Figs. 5 and 6.

$$\Delta I_{HV \ side} = \frac{V_i}{fL} D \tag{9}$$

$$\Delta V_{HV \ side} = \frac{I_o}{fC} D \tag{10}$$

$$\Delta I_{LV \ side} = \frac{V_o \left( V_i - V_o \right)}{f L \ V_i} \ D \tag{11}$$

$$\Delta V_{LV \ side} = \frac{V_i}{8 L C f^2} D (1-D)$$
(12)

where:

 $\Delta I_{HV}$ , the inductor current for the boost side;

## Table 7

BESS systems for the MG distribution system.

ID	Battery Unit	Nominal Voltage (V)	Rated Capacity (Ah)	Initial SOC (%)
BESS # 1	Battery 1	120	800	80
BESS #	Battery 1	650	1,5	80
2	Battery 2	650	1,5	80
	Battery 3	650	1,5	80

 $\Delta V_{HV}$ , the capacitor voltage ripple for the boost side;  $\Delta I_{LV}$ , is the inductor current for the buck side;  $\Delta V_{LV}$ , is the capacitor voltage ripple for the buck side.

There is no doubt that bidirectional AC/DC converters have many applications in the microgrid field. Therefore, the power interaction between the DC bus and the AC bus (see Fig. 7), was proposed in this study using two bidirectional converters of several switching states for the inversion or rectifier modes. For this conversion system, a threephase universal power module is implemented. It consists of six power switches connected in a bridge configuration, by means of electronic power devices such as diodes and IGBTS (device-controlled switching).

A configuration of two-paralleled bidirectional AC/DC converters is presented. It is designed to convert the voltage from 300 VDC to an output voltage of 150 VAC. It allows to exchange the power generated in the DC bus to the AC grid. This solution becomes interesting from the point of view of increasing robustness, flexibility, and performance of the Microgrid in DC, which are designed to maximize performance for a wide range of powers. The bidirectional converters were considered with a master-slave control scheme.

The input voltage ( $V_{input}$ ) and output voltage ( $V_{output}$ ) are the same for the two conversion modules, just as the input current ( $I_{input}$ ) and output current ( $I_{output}$ ) are the result in the sum of the currents in each individual module, see Eqs. (13) and (14).

So,

$$I_{input} = \sum_{j=1}^{n} I_{input, j} \text{ and } I_{output} = \sum_{j=1}^{n} I_{output, j}$$
(13)

$$P_{input} = \sum_{j=1}^{n} P_{input, j} \text{ and } P_{output} = \sum_{j=1}^{n} P_{output, j}$$
(14)

where:

*n* is the total number of modules (for this case n = 2);  $I_{input,j}$  is the input current of the jth module;  $P_{input,j}$  is the input power of the jth module.

### 4. Power flow results

The AC/DC MG test system model as shown in Fig. 4, is simulated

using Simscape within Simulink. In order to obtain a better software performance during the simulation process, a step time of 5 µs is chosen. Values for all angle buses are computed by using a PLL-based adaptive notch filter [59]. The Phase-Locked Loop auxiliary circuit extracts the phase angle from the instantaneous voltage waveform of one phase as well as the phase angle from the instantaneous current waveform of the same phase. Then, a subtraction is made of these two values to compute  $\varphi$  for every bus. Further details of the PLL design and its performance are shown in [59].

The power block computes the active power  $(P_i)$  and the reactive power  $(Q_i)$  of a voltage-current pair signal at the fundamental frequency. In order to perform this computation, the block determines the peak magnitude of the two input signals  $V_i(t) \& I_i(t)$ ,  $|V_i|$  and  $|I_i|$ . The output variables  $(P_i)$  and  $(Q_i)$  are then calculated in every period of the fundamental waveform, as shown in Eqs. (15), (16), and (17):

$$\varphi_i = \angle V_i - \angle I_i \tag{15}$$

$$P_i = \frac{|V_i|}{\sqrt{2}} \frac{|I_i|}{\sqrt{2}} \cos(\varphi_i)$$
(16)

$$Q_i = \frac{|V_i|}{\sqrt{2}} \frac{|I_i|}{\sqrt{2}} \sin(\varphi_i)$$
(17)

where:

Table 8

 $\left|V_{i}\right|$  is the peak voltage value for one phase;

 $\left|I_{i}\right|$  is the peak current value for the same phase;

 $\phi_i$  is the difference between the voltage angle and current angle for the same phase;

P<sub>i</sub> is the per-phase active power;

Q<sub>i</sub> is the per-phase reactive power;

*i* is the bus number.

This section reports on the simulation results for the proposed test system. The MG keeps its operation stable at steady state for both conditions: maximum and minimum demand scenarios. Although the calculations furnished by Simulink are per phase, active and reactive power values shown on Tables 8 and 9 are the sum of the per-phase power that is, three-phase active power and reactive power, respectively. For any generator bus,  $P_g$  and  $Q_g$  are presented, whereas for any load bus,  $P_1$  and  $Q_1$  are presented. Lastly, any transfer bus,  $P_{transf}$  and  $Q_{transf}$  are shown.

## 5. Simulation analysis of the proposed model

In this section, the reported results are analyzed and metrics of the quality and efficiency variables of the power will also be established. To compute metrics that allow defining a base case for future optimization and compensation studies, two analysis scenarios are defined.

The daily demand curve of the proposed MG is shown in Fig. 8, which is obtained as the sum of the hourly demands of each load of the model with predominant components of industrial and commercial loads. The loads were modelled with constant impedance and for this reason, a slight difference between the load nominal power and the actual power consumed can be determined. This is due to the voltage drops in the system.

In the daily total demand curve shown in Fig. 8, the minimum demand of the proposed system is approximately 30% of the maximum demand, and this same relation between the active power demands was considered in the total reactive power of the system.

This remarkable difference between the two scenarios allows a real

Bus i	Туре	P <sub>g</sub> (kW)	Q <sub>g</sub> (kVAr)	P <sub>l</sub> (kW)	Q <sub>l</sub> (kVAr)	P <sub>transf</sub> (kW)	Q <sub>transf</sub> (kVAr)	V (pu)	δ (°)
LV 1	BESS 2	42,66	30,45	-	-	-	-	0,955	-29,76
LV 2	Transfer Bus	-	-	-	-	41,38	4,91	0,931	-30,76
LV 3	Transfer Bus	-	-	64,72	40,11	62,6	43,86	0,930	-31,26
LV 4	Transfer Bus	-	-	-	-	120,66	84,61	0,953	-31,5
LV 5	Transfer Bus	-	-	-	-	34,73	58,77	0,951	-31,25
MV 6	Transfer Bus	-	-	-	-	780	1095	0,966	-30,31
MV 7	Transfer Bus	-	-	-	-	554,6	356,2	0,971	-30,72
MV 8	Diesel	690	450	0	0	-	-	0,975	-60,84
MV 9	Non-linear load	-	-	327,3	38,23	-	-	0,966	-30,67
MV 10	-	0	0	572,4	427,2	-	-	0,94	-29,81
MV 11	-	-	-	290,28	217,71	-	-	0,953	-30,01
MV 12	-	-	-	586,2	439,8	-	-	0,957	-30,11
MV 13	Slack	1810,2	1665	0	0	-	-	0,974	-30,66
MV 14	Transfer Bus	-	-	119,61	89,7	226,02	397,2	0,967	-30,59
DC	DC Bus	-	-	-	-	8	0	0,928	0

Table 9

Load flow results for the minimum demand scenario.

Bus	Туре	P <sub>g</sub> (kW)	Q <sub>g</sub> (kVAr)	P <sub>1</sub> (kW)	Q <sub>l</sub> (kVAr)	P <sub>transf</sub> (kW)	Q <sub>transf</sub> (kVAr)	V (pu)	δ (°)
LV 1	BESS 2	27,6	15,75	-	-	-	-	0,9567	-31,27
LV 2	Transfer Bus	-	-	-	-	12,64	5,77	0,959	-31,28
LV 3	Transfer Bus	-	-	21,1	13,09	25,47	14,43	0,967	-31,14
LV 4	Transfer Bus	-	-	-	-	50,3	25,74	0,979	-31,14
LV 5	Transfer Bus	-	-	-	-	51	26,79	0,983	-31,24
MV 6	Transfer Bus	-	-	-	-	471,2	346,1	0,980	-30,31
MV 7	Transfer Bus	-	-	-	-	89	84,4	0,98	-30,25
MV 8	Diesel	150,1	49	0	0	-	-	0,98	-60,23
MV 9	Non-linear load	-	-	110,2	0,69	-	-	0,98	-30,28
MV 10	-	0	0	183	135	-	-	0,97	-30,15
MV 11	-	-	-	92,1	69	-	-	0,97	-30,21
MV 12	-	-	-	184,8	138,6	-	-	0,98	-30,24
MV 13	Slack	91,8	74,1	0	0	-	-	0,99	-30,39
MV 14	Transfer Bus	-	-	372	279	21,84	84,6	0,98	-30,3
DC	DC Bus	-	-	-	-	0,975	0	0,985	0



Fig. 8. Demand curve. As previously stated, the demand for electrical energy was defined and modelled for two scenarios for which the following dynamic points of operation shown in Table 10 were taken into account.

#### Table 10

Generations modes for both demand scenarios.

Generations	Maximum demand	Minimum demand
Main Grid	Connected	Connected
Solar PV Array 1 & 2	Generating at 1000 W/m <sup>2</sup>	OFF at 0 W/m <sup>2</sup>
BESS 1	Charge at 5A	Charge at 3A
BESS 2	Generated	Charge
Bidirectional Converter 1.1 & 1.2	Inverter	Rectifier
Diesel Generator	Generating	Generating

future analysis to be established in terms of reactive power compensation. The capacity and location of the compensating devices must be determined considering the minimum demand scenario. In fact, the voltage profile of the system could reach overvoltage levels with the reactive injection of the compensating devices.

In Fig. 8, it can also be seen that the maximum demand period is at 11h00, which coincides with the peak of solar radiation and, therefore, with the maximum generation capacity of the solar panels. This demand peak during this time is due to the integration of industrial load curves with residential load curves.

#### 5.1. Voltage profile analysis

Figs. 9 and 10 show the voltage profile results obtained in the scenarios of maximum load power analysis (Fig. 9) and minimum load power (Fig. 10). The analysis of the voltage profiles is shown in phases because the system presents unbalanced loads as a result of single-phase load components. In both cases, it can be seen that the voltage profiles are in permissible operating parameters; however, in maximum demand (Fig. 9) there are significant voltage drops in the low voltage grid (220 V) in buses 2 y 3. This result allows future study for optimal compensation of this indicator. Both figures show indicators that measure the quality of the voltage profile, and the calculation methods used are detailed in Eqs. (18) and (19).

Average deviation of the voltage in the system.

$$ADVS = \frac{\sum_{i=1}^{n} |Vd_i - V_i|}{n}$$
(18)

Maximum value of voltage deviation.

This function represents the maximum voltage deviation in the system.

$$MVD = \max_{1 \le i \le n} (|Vd_i - V_i|)$$
(19)

where:

*n* is the number of buses in the MG;

 $V_i$  is the real voltage at bus *i* in p.u. (per unit);

 $Vd_i$  is the desired voltage at bus *i* in p.u. (1 p.u.);

**ADVS** is a figure of merit that is required to be minimal in a future optimization problem for the improvement of voltage profiles.

Fig. 9 shows that the maximum value of the average deviation of the system voltage in the maximum demand analysis does not exceed 0,046 per unit. However, the maximum value of voltage deviation (MVD) in this system is 0,0691 (phase a = 0,0859; phase b = 0,0492; phase c = 0,0721), and it is in bus 3. This result evidences the need to propose future studies based on finding solutions for the improvement of the voltage profile in this system.

Fig. 10 shows that the voltage profile in the minimum demand scenario presents positive results for all the buses. The average deviation calculations of the system voltage and maximum deviation are very low, which indicates an operation within the desired parameters. This scenario was calculated and analyzed to be considered in reactive power compensation proposals to solve the maximum demand scenario that could affect the voltage profile in the minimum demand scenario.

### 5.2. Active power balance per bus

Figs. 11 and 12 show the contribution of the powers in kVA generated and consumed in each of the buses of the system, including the kW power on the DC bus. The positive graphical powers indicate load powers connected in that bus or powers transferred, except in the case of bus 14 in which a negative power is shown, and it is transferred from bus 9 to bus 14 as a power supply contribution of the Diesel generator (this explanation is detailed in Tables 8 and 9) and the rest of negative powers indicate generation in the bus to which corresponds.









In the scenario of maximum demand (Fig. 11) there is a contribution of generation power in 6 buses (1, 2, 6, 8, 13 and DC Bus).

Fig. 12 shows the power balance of the system in the minimum demand scenario, in which the power decreases in each bus with respect to the maximum demand scenario. The contribution in distributed generation for this scenario is also shown, leaving only the Diesel generator connected at bus 8 and the rest of the contribution in active power is delivered by the connected grid through bus 13.

## 5.3. Reactive power balance per bus and angular analysis

Figs. 13 and 14 show the reactive power balance per bus, the voltage







Fig. 12. Power balance in minimum demand.

profile per bus and the variation of phase shift in the voltage angles at the system buses, in maximum and minimum demand respectively. It can be seen that in both figures, there is a coherent relation between the reactive injection and the increase in the voltage profile at the bus where the reactive power is compensated with the existing generation in the

### system.

The contribution of reactive power in this system is found in bus 8 with the diesel generator and in bus 13 with the reactive power contribution of the external grid. However, in maximum demand (Fig. 13) an apparent injection of reactive power can be seen in buses 9 and 14.



Fig. 13. Analysis per bus and phase, in maximum demand scenario, (a) reactive power balance, (b) voltage profile and (c) angular analysis.



Fig. 14. Analysis per bus and phase in minimum demand scenario, (a) reactive power balance, (b) voltage profile and (c) angular analysis.

Transfer buses show negative powers due to the direction of power flows in this scenario. This explanation is detailed in Tables 8 and 9.

These scenarios provide an important starting point for the analysis of reactive power compensation in order to improve the voltage profiles. The contribution of a minimum demand analysis allows to validate a compensation result in the maximum demand scenario.

Fig. 14 shows the voltage profile in this scenario. It has good values and in some buses, it exceeds 1 p.u. This situation needs a analysis when raising the reactive injection to improve the maximum demand scenario.

#### 5.4. Power losses per line

The losses of active power in this system were calculated under conditions of balance and unbalance of currents as defined in [60]. The power losses for a state of charge a on a line b ( $\Delta P_{a,b}$ ), are as indicated in Eq. (20):

$$\Delta \boldsymbol{P}_{a,b} = \boldsymbol{R}_{I} * \sum_{i=1}^{Nf} \left| \boldsymbol{I}_{i,a,b} \right|^{2} + \boldsymbol{R}_{n} * \sum_{i=1}^{Nf} \left| \boldsymbol{I}_{i,a,b} \right|^{2}$$
(20)

where:

 $I_{i,a,b}$  is the current in phase *i* during the state of charge *a*, circulating in line *b*;

 $R_l$ ,  $R_n$  are the resistors of the phase and neutral conductors respectively;

*Nf* is the number of phases of the system in line *b*.

Figs. 15 and 16 show the losses per phase in each of the lines of the system, in the scenarios of maximum demand and minimum demand respectively. Both figures show that the highest loss levels correspond to the lines that feed buses in the low voltage grid. It can be seen that the highest level of losses per phase (of almost 9.9 kW) is presented in phase a of line 7 (connection between buses 3 and 4), a value that corresponds

to the transport of high unbalanced powers in this line, even though its distance is relatively short.

Figs. 15 and 16 show the unbalance of active power losses per phase that exists in the buses with unbalanced loads. This unbalance is due to the component of single-phase loads connected in the low voltage grid of this system.

Fig. 16 shows that the highest level of losses per phase does not exceed 1,4 kW, which is a very low value with respect to the maximum demand losses per phase.

#### 5.5. Power factor per bus

The power factor per bus is one of the most important variables that should be considered in the calculation of power flows in systems which need to be compensated. Figs. 17 and 19 show the power factor for each bus of the system in the scenarios of maximum demand and minimum demand respectively. In the scenario of maximum demand (Fig. 17) it can be seen that the power factor in buses 2 and 9 is almost equal to 1. This is due to the fact that in Bus 2, in maximum demand, there is only a contribution of active power from the DC MG, while the almost unitary power factor in bus 9 is due to the fact that the measurement was made directly in the non-linear load, as shown in Fig. 18. Tables 11 and 12 show the relation between the active and reactive powers in bus 9 corresponding to the non-linear load, in the maximum and minimum demand scenarios.

In both scenarios, it demonstrates that there is a great deterioration of the power factor in bus 14 because this bus is a transfer bus as shown in Fig. 18. Tables 11 and 12 show the results of the power flows in bus 14, where it can be seen that the measurement of flows in this bus contains the flows of load 14 and the transfer flow to line 14, which is transporting a reactive power higher than the active power, as shown in Tables 11 and 12 for each scenario. A similar situation occurs in bus 5 as it only operates as a transfer bus, connecting line 2, line 3 and line 5. This bus is also transporting a reactive power higher than the active power. BESS #2



Fig. 15. Power losses per line in maximum demand.



Fig. 16. Power losses per line in minimum demand.



Fig. 17. Power factor per bus in maximum demand.

alone is operating at 0.8 lagging.

The comparison of power factors between the two scenarios shows some results that are particularly important. For example, in conditions of minimum demand, the power factor is much better in most of the buses than the power factor in the maximum demand scenario. The relation of decrease in the reactive power of the maximum demand scenario to the minimum demand scenario is the same relation of decrease for reactive power. With this analysis it might seem that the power factor seen by the grid should be the same. However, the power factor in maximum demand is 0,73 while in minimum demand, it is 0,77.

This remarkable difference between the power factors in both scenarios can also be seen in buses 5 and 6. This phenomenon is due to the



Fig. 18. Detail of MATLAB /Simulink topology of bus 9 and 14.

Table 11Maximum demand power factor for bus 14 and 9: Detail.

Bus	Details	P (kW)	Q (kW)	PF
MV bus 14	Bus 14_1	970	499,8	0,88
	Load 14	1196,1	897	0,8
	MV Bus 14	226,02	397,2	0,49
MV bus 9	MV bus 9	327,3	38,23	0,99

Table 12

|--|

Bus	Details	P (kW)	Q (kW)	PF
MV bus 14	Bus 14_1	393,84	363,6	0,73
	Load 14	372	279	0,8
	MV Bus 14	21,84	84,6	0,23
MV bus 9	MV bus 9	110,2	0,69	1

loss of active power contribution in the solar photovoltaic generations, which do not occur in the scenario of minimum demand that takes place at night when there is no solar radiation. This analysis shows the conflict that can exist in the variable power factor when it comes to compensating a power electrical system only with the contribution of active power. With the injection of only active power, the power factor seen by the grid deteriorates, which leads to the need to compensate a system with more criteria and where joint compensation of active and reactive power is considered.

The power factor in bus 7 in the scenario of minimum demand shown

in Fig. 19 is at a lower value with respect to the power factor in the same bus in maximum demand, this is due to the contribution of reactive power of the Diesel generator depending on the power flows.

## 5.6. THD per bus

This section presents the analysis of the deformation of the sine wave voltage. This is one of the problems that affects the quality of electric power especially in distribution systems. Figs. 20 and 21 show the Harmonic Distortion Index (THD) of the voltage for each of the buses of the system and in each of the phases, in the scenarios of maximum demand and minimum demand, respectively. It can be seen in both figures that the harmonic distortion does not exceed 3,5 % for any of the calculated scenarios. However, these values are a base case for future studies with insertion of non-linear loads and for optimization problems that involve the improvement of THD and the compensation of the reactive by means of equipment based on power electronics. The THD in each bus is calculated as shown in Eq. (21) [61, 62, 63].

$$THD\% = \frac{\sqrt{\sum_{i=2}^{H} (V_{i,h})^2}}{V_{i,1}} * 100$$
(21)

where:

 $V_{i,h}$  is the voltage component corresponding to the harmonic *h* at the node *i*;







Fig. 20. THD per bus in maximum demand.

 $V_i, 1$  is the fundamental voltage component (1<sup>st</sup> harmonic) at the node *i*;

H, is the maximum harmonic order to be considered in the calculation.

## 6. Future work and research areas

As in [4], international organizations such as Power Systems Engineering Research Center (PSERC), Consortium for Electric Reliability Technology Solutions (CERTS), Consortium for Electric Reliability Technology Solutions (NEDO), Transmission and Distribution Research Center (T&D) among others, have presented research projects in Smart



Fig. 21. THD per bus in minimum demand.

MGs to the scientific community. All these entities have the aim of guaranteeing the speed of the grid and the quality of the energy, integration of resources and distributed generation, generation of active Distribution Grids, emissions reduction, decrease of construction and investment times, and reduction of losses in T&D.

In addition, the nature of modern power grids has gone from being dominated by a few generation units to a grid dominated by many smaller generating units, together bringing a growing number of renewable generation sources such as wind turbines, photovoltaic systems, and energy storage systems. This has led to great research efforts in power grids and smart MGs to mitigate new challenges such as: low and erratic consumption, non-existent decoupling of P - f - |V|; the loss of coordination of MG protection; voltage unbalance conditions; increased voltage; the connection/disconnection of DG sources; high failure rates, among others.

As it has been commented in the introduction, some of the interfacing converters are in an open control loop, i.e. no control over the index modulation nor over the phase angle has been implemented. In a previous study, this technique was successfully used to establish a certain operation point [64]. This is a pending issue for the benchmark [65]. There are other tests for the benchmark [66, 67].

This study offers the possibility of developing in research areas that must be consider:

Grid-islanded mode [44, 68], synchronization modes to the Main Grid [24, 69], tests over the MG controllability in the islanded configuration [70], MG modelling, and analysis [10, 47, 71, 72, 73], tests over reactive compensation as a solution for single phase non-linear loads like electric arc furnaces using Static Var Compensation (SVC) [29], tests over power quality issues. Explore the possibility of using Distribution STACOM, Green Plug Filter Compensators (GPFC) or Modulated Power Filter Compensators (MPFC) [2, 74], tests over optimization and sizing techniques [75, 76, 77], control of power electronic converters [DC MG Control (boost-buck)] and hierarchical control structure tests [1, 21, 30, 38, 78, 79, 80, 81, 82, 83, 84].

Another critical problem in MGs is the growth in the number of failures in their components, especially the failures that comprise the infrastructure of control and automation, communications, generation, and distribution of energy.

In the future, fault modeling and analysis may be performed which not only allows us to understand the various failure modes (their causes and effects) but also to study optimal methodologies for taking timely detection, diagnosis and localization actions.

The objective is to develop and sustain the development of real-time automated tool research that allows mitigation and failure relief [85], the development of fault-tolerant control strategies [78, 86, 87, 88, 89, 90, 91, 92, 93, 94, 95, 96, 97, 98, 99, 100, 101, 102, 103], intelligent and adaptive protections in AC and DC [6, 104], among others. The support of these projects such as those described in [85, 86, 93], confirms and shows dynamic behaviors that benefit engineers that work in the area of fault diagnosis as well as those that develop fault tolerance methodologies for control, which in turn helps to improve the resilience and reliability of the MGs. Some development in predictive [105, 106, 107, 108, 109] and hierarchical control techniques [20, 38, 40, 80, 82, 110, 111, 112, 113] can be explored in this approach, among others. The support of these projects such as those described in [85, 86, 93], shows dynamic behaviors that benefit engineers that work in the area of fault diagnosis as well as those that develop fault tolerance methodologies for control. This in turn helps to improve the resilience and reliability of the MGs. Some developments in predictive [109, 110, 111, 112, 113] and hierarchical control techniques [20, 38, 40, 80, 82, 105, 106, 107, 108] can be explored in this approach.

In addition, the base case obtained in this research with maximum and minimum demand scenarios allows future studies in many areas of great current interest, associated with the efficiency, reliability and quality of the electric power. The proposed system allows the analysis of power dispatch in the event of grid contingencies or unbalances between load and generation. The impact of the insertion of the electric car in the demand studies of both scenarios can be analyzed.

This model may be the base for future studies such as optimal dimension and location of reactive power compensating devices. It is also possible to analyze the reliability indicators and grid expansion proposals for future load growth. In addition, dynamic analysis of the stability of the frequency could be carried out for disturbances, faults and unbalances between generation and load, especially considering the loss of inertia in the system with isolated mode.

#### 7. Conclusions

In this paper, a typical AC/DC HMG benchmark based on the IEEE 14 node test feeder is presented. The selection of renewable energy sources have considered the availability of photovoltaic energy resources instead of wind energy resources. This AC/DC HMG benchmark includes a oneline diagram as well as essential data for the 13,8-kV primary system and 0,22-kV secondary system. The proposed study is subjected to two scenarios: minimum and maximum power demands. Both power flow results are published for each scenario. All buses have their measurement module. Simulation of measurements related with power quality indexes as current and voltage total harmonic distortion (THDV and THDI) are calculated, including capability to compute power line losses and power factor. The results obtained in the analysis of the two scenarios calculated in this proposed model have been discussed individually, allowing a wide understanding of the problem of each of the variables involved in the analysis of efficiency and quality of the power. The conflict that exists in some of the variables was analyzed, especially in the power factor, due to the high penetration of photovoltaic solar generation, which can cause a deterioration of the power factor seen by the commercial entities that provide the electric service in the mode connected to the grid.

In the future, MGs are an attractive solution for the integration of DG units in the Smart Grid and may help lessen a dependence on fossil fuels and increase the efficiency of the electrical grid. However, the challenges are still present due to the following factors: the rapid dynamics and short response time of DG or distributed energy resources, the inherent unbalanced nature of the MG, low energy storage capacity and lack of inertia, a high number and diversity of micro sources used, electronic device power converters and other circuits/devices, a high degree of parametric uncertainties, modelling, and high failure rates.

This benchmark study opens the possibility to investigate transient stability, test control strategies and hierarchical control structure, explores isolated scenarios, simulates the dynamics of the diesel generator, compensates voltage profiles, among others. The investigation allows to deepen the scenarios and topologies of the interconnected AC/DC HMGs and allows dynamic studies for other AC/DC HMG configurations, even though it depends on a large extent on the application and the integrated environment (market conditions, regulatory conditions, feasibility for the integration of DG units, among others). Finally, this study constitutes a base to develop research in the framework of MGs, its potential applications, as well as improve reliability, efficiency and device costs.

## Declarations

## Author contribution statement

Leony Ortiz, Rogelio Orizondo & Alexander Águila: Conceived and designed the experiments; Performed the experiments; Analyzed and interpreted the data; Contributed reagents, materials, analysis tools or data; Wrote the paper.

Jorge W. González, Gabriel J. López & Idi Isaac: Conceived and designed the experiments; Analyzed and interpreted the data; Contributed reagents, materials, analysis tools or data; Wrote the paper.

## Funding statement

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

#### Competing interest statement

The authors declare no conflict of interest.

## Additional information

No additional information is available for this paper.

## References

- [1] D.E. Olivares, A. Mehrizi-Sani, A.H. Etemadi, C.A. Cañizares, R. Iravani, M. Kazerani, A.H. Hajimiragha, O. Gomis-Bellmunt, M. Saeedifard, R. Palma-Behnke, G.A. Jiménez-Estévez, N.D. Hatziargyriou, Trends in microgrid control, IEEE Trans. Smart Grid. 5 (2014) 1905–1919.
- [2] H.A. Gabbar, Smart Energy Grid Engineering, Joe Hayton, Canada, 2017.
- [3] A.K.G.K.G. Dimeas Aris, Tsikalakis, Microgrids: Architectures and Control, 2014.
- [4] E. Planas, J. Andreu, J.I. Gárate, I. Martínez De Alegría, E. Ibarra, AC and DC technology in microgrids: a review, Renew. Sustain. Energy Rev. 43 (2015) 726–749.
- [5] L. Jia, Y. Zhu, Y. Wang, Architecture design for new AC-DC hybrid micro-grid, in: IEEE 1st Int. Conf. Direct Curr. Microgrids, ICDCM 2015, 2015, pp. 113–118.
- [6] S. Parhizi, H. Lotfi, A. Khodaei, S. Bahramirad, State of the art in research on microgrids: a review, IEEE Access 3 (2015) 890–925.
- [7] F. Martin-Martínez, A. Sánchez-Miralles, M. Rivier, A literature review of Microgrids: a functional layer based classification, Renew. Sustain. Energy Rev. 62 (2016) 1133–1153.
- [8] E. Hossain, E. Kabalci, R. Bayindir, R. Perez, Microgrid testbeds around the world: state of art, Energy Convers. Manag. 86 (2014) 132–153.
- [9] G. Turner, J.P. Kelley, S. Member, C.L. Storm, D.A. Wetz, S. Member, W. Lee, Design and active control of a microgrid testbed, IEEE Trans. Smart Grid. 6 (2015) 73–81.
- [10] M.S. Mahmoud, S.A. Hussain, M.A. Abido, Modeling and control of microgrid : an overview, J. Frankl. Inst. 351 (2014) 2822–2859.
- [11] R.H. Lasseter, J.H. Eto, B. Schenkman, J. Stevens, H. Vollkommer, D. Klapp, E. Linton, H. Hurtado, J. Roy, CERTS microgrid laboratory test bed, IEEE Trans. Power Deliv. 26 (2011) 325–332.
- [12] N.W.A. Lidula, A.D. Rajapakse, Microgrids research: a review of experimental microgrids and test systems, Renew. Sustain. Energy Rev. 15 (2011) 186–202.
- [13] P. Piagi, R.H. Lasseter, Autonomous control of microgrids, in: 2006 IEEE Power Eng. Soc. Gen. Meet., 2006, p. 8.
- [14] E. Commission, Smart Grid Projects Map: Organisations and Implementation Sites, 2018. http://ses.jrc.ec.europa.eu/.
- [15] E.D.& E. RELIABILITY, Microgrid Portfolio of Activities, 2012. https://www.energ y.gov/oe/services/technology-development/smart-grid/role-microgrids-helping-a dvance-nation-s-energy-syst-0.
- [16] A. Kaur, J. Kaushal, P. Basak, A review on microgrid central controller, Renew. Sustain. Energy Rev. 55 (2016) 338–345.
- [17] M.H. Andishgar, E. Gholipour, R. allah Hooshmand, An overview of control approaches of inverter-based microgrids in islanding mode of operation, Renew. Sustain. Energy Rev. 80 (2017) 1043–1060.
- [18] Y. Yoldaş, A. Onen, S.M. Muyeen, A.V. Vasilakos, İ. Alan, Enhancing smart grid with microgrids: challenges and opportunities, Renew. Sustain. Energy Rev. 72 (2017) 205–214.
- [19] J. John, F. Mwasilu, J. Lee, J. Jung, AC-microgrids versus DC-microgrids with distributed energy resources : a review, Renew. Sustain. Energy Rev. 24 (2013) 387–405.
- [20] E. Unamuno, J.A. Barrena, Hybrid ac/dc microgrids Part I: review and classification of topologies, Renew. Sustain. Energy Rev. 52 (2015) 1251–1259.
- [21] E. Unamuno, J.A. Barrena, Hybrid ac/dc microgrids Part II: review and classification of control strategies, Renew. Sustain. Energy Rev. 52 (2015) 1123–1134.
- [22] S. Mirsaeidi, X. Dong, D.M. Said, Towards hybrid AC/DC microgrids: critical analysis and classification of protection strategies, Renew. Sustain. Energy Rev. 90 (2018) 97–103.
- [23] A. Hirsch, Y. Parag, J. Guerrero, Microgrids: a review of technologies, key drivers, and outstanding issues, Renew. Sustain. Energy Rev. 90 (2018) 402–411.
- [24] M.R. Amin, S. Aizam Zulkifli, A framework for selection of grid-inverter synchronisation unit: harmonics, phase-angle and frequency, Renew. Sustain. Energy Rev. 78 (2017) 210–219.
- [25] R. Ortega, E. Figueres, G. Garcerá, C.L. Trujillo, D. Velasco, Control techniques for reduction of the total harmonic distortion in voltage applied to a single-phase inverter with nonlinear loads: Review, Renew. Sustain. Energy Rev. 16 (2012) 1754–1761.
- [26] P. Basak, S. Chowdhury, S. Halder Nee Dey, S.P. Chowdhury, A literature review on integration of distributed energy resources in the perspective of control, protection and stability of microgrid, Renew. Sustain. Energy Rev. 16 (2012) 5545–5556.
- [27] U.B. Tayab, M.A. Bin Roslan, L.J. Hwai, M. Kashif, A review of droop control techniques for microgrid, Renew. Sustain. Energy Rev. 76 (2017) 717–727.
- [28] M.R. Miveh, M.F. Rahmat, A.A. Ghadimi, M.W. Mustafa, Control techniques for three-phase four-leg voltage source inverters in autonomous microgrids: a review, Renew. Sustain. Energy Rev. 54 (2016) 1592–1610.
- [29] M.T.L. Gayatri, A.M. Parimi, A.V. Pavan Kumar, A review of reactive power compensation techniques in microgrids, Renew. Sustain. Energy Rev. 81 (2018) 1030–1036.
- [30] Z. Shuai, J. Fang, F. Ning, Z.J. Shen, Hierarchical structure and bus voltage control of DC microgrid, Renew. Sustain. Energy Rev. 82 (2018) 3670–3682.

- [31] S.R. Sivarasu, E. Chandira Sekaran, P. Karthik, Development of renewable energy based microgrid project implementations for residential consumers in India: scope, challenges and possibilities, Renew. Sustain. Energy Rev. 50 (2015) 256–269.
- [32] A. López-González, B. Domenech, D. Gómez-Hernández, L. Ferrer-Martí, Renewable microgrid projects for autonomous small-scale electrification in Andean countries, Renew. Sustain. Energy Rev. 79 (2017) 1255–1265.
- [33] Z. Xu, P. Yang, C. Zheng, Y. Zhang, J. Peng, Z. Zeng, Analysis on the organization and Development of multi-microgrids, Renew. Sustain. Energy Rev. 81 (2018) 2204–2216.
- [34] E.E. Gaona, C.L. Trujillo, J.A. Guacaneme, Rural microgrids and its potential application in Colombia, Renew. Sustain. Energy Rev. 51 (2015) 125–137.
- [35] C. Phurailatpam, B.S. Rajpurohit, L. Wang, Planning and optimization of autonomous DC microgrids for rural and urban applications in India, Renew. Sustain. Energy Rev. 82 (2018) 194–204.
- [36] L. He, S. Zhang, Y. Chen, L. Ren, J. Li, Techno-economic potential of a renewable energy-based microgrid system for a sustainable large-scale residential community in Beijing, China, Renew. Sustain. Energy Rev. 93 (2018) 631–641.
- [37] M. Wolsink, The research agenda on social acceptance of distributed generation in smart grids: renewable as common pool resources, Renew. Sustain. Energy Rev. 16 (2012) 822–835.
- [38] J.M. Guerrero, J.C. Vasquez, J. Matas, L.G. De Vicuña, M. Castilla, Hierarchical control of droop-controlled AC and DC microgrids - a general approach toward standardization, IEEE Trans. Ind. Electron. 58 (2011) 158–172.
- [39] B. Kroposki, E.D. Anese, A. Bernstein, Y. Zhang, B. Hodge, Autonomous energy grids preprint, in: Hawaii Int. Conf. Syst. Sci., 2018.
- [40] Y. Gao, Q. Ai, Distributed cooperative optimal control architecture for AC microgrid with renewable generation and storage, Int. J. Electr. Power Energy Syst. 96 (2018) 324–334.
- [41] T. Samad, A. Annaswamy, Controls for smart Grids : architectures and applications, in: Proc. IEEE., 2017, pp. 1–18.
- [42] E. Prieto-araujo, D.B. Bolboceanu, S. Enric, Design methodology of the primary droop voltage control for DC microgrids, DC Microgrids (ICDCM), in: 2017 IEEE Second Int. Conf., 2017.
- [43] M.S. Mahmoud, N.M. Alyazidi, M.I. Abouheaf, Adaptive intelligent techniques for microgrid control systems: a survey, Int. J. Electr. Power Energy Syst. 90 (2017) 292–305.
- [44] M. Sadegh, G. Esfahani, Coordinated Control of Distributed Energy Resources in Islanded Microgrids, 2016.
- [45] A.H. Fathima, K. Palanisamy, Optimization in microgrids with hybrid energy systems - a review, Renew. Sustain. Energy Rev. 45 (2015) 431–446.
- [46] C. Gamarra, J.M. Guerrero, Computational optimization techniques applied to microgrids planning: a review, Renew. Sustain. Energy Rev. 48 (2015) 413–424.
- [47] A.M. Vega, F. Santamaria, E. Rivas, Modeling for home electric energy management: a review, Renew. Sustain. Energy Rev. 52 (2015) 948–959.
- [48] R.H. Lasseter, P. Paigi, Microgrid : a conceptual solution, in: Annu. IEEE Power Electron. Spec. Conf., 2004, pp. 4285–4290.
- [49] Hassan Bevrani, T. Ise, Others, Microgrid Dynamics and Modeling, John Wiley \& Sons, 2017.
- [50] N. Bottrell, T.C. Green, Modeling microgrids with active loads, in: IEEE 13th Work. Control Model. Power Electron. COMPEL 2012, 2012, 2012.
- [51] X. Tang, W. Deng, Z. Qi, Investigation of the dynamic stability of microgrid, IEEE Trans. Power Syst. 29 (2014) 698–706.
- [52] A.H. Etemadi, R. Iravani, Eigenvalue and robustness analysis of a decentralized voltage control scheme for an islanded multi-DER microgrid, IEEE Power Energy Soc. Gen. Meet (2012) 1–8.
- [53] O. Tremblay, L.A. Dessaint, Experimental validation of a battery dynamic model for EV applications, World Electr. Veh. J. 3 (2009) 289–298.
- [54] D.W. Gao. Energy Storage for Sustainable Microgrid Energy Storage for Sustainable Microgrid, 1st edition, Academic Press, USA, 2015.
- [55] A. Delavari, I. Kamwa, A. Zabihinejad, A comparative study of different multilevel converter topologies for battery energy storage application, in: 2017 IEEE 30th Can. Conf. Electr. Comput. Eng. A, Windsor, ON, Canada, 2017, pp. 861–865.
- [56] R. Viral, D.K. Khatod, Optimal planning of distributed generation systems in distribution system: a review, Renew. Sustain. Energy Rev. 16 (2012) 5146–5165.
- [57] M. Saleh, Y. Esa, Y. Mhandi, W. Brandauer, A. Mohamed, Design and implementation of CCNY DC microgrid testbed, in: IEEE Ind. Appl. Soc. 52nd Annu. Meet. IAS 2016, 2016, pp. 1–7.
- [58] R. Abid, F. Masmoudi, F. Ben Salem, N. Derbel, Modeling and simulation of conventional DC-DC converters deticated to photovoltaic applications, in: IREC 2016 - 7th Int. Renew. Energy Congr., 2016, pp. 1–6.
- [59] M. Karimi-Ghartemani, M.R. Iravani, A nonlinear adaptive filter for online signal analysis in power systems: Applications, IEEE Trans. Power Deliv. 17 (2002) 617–622.
- [60] A. Aguila, D. Carrión, L. Ortiz, Analysis of power losses in the asymmetric construction of electric distribution systems, IEE Lat. Am. Trans. 13 (2015) 2190–2194.
- [61] C.-S. Lam, L. Wang, S.-I. Ho, M.-C. Wong, Adaptive thyristor-controlled LC-hybrid active power filter for reactive power and current harmonics compensation with switching loss reduction, IEEE Trans. Power Electron. 32 (2017) 7577–7590.
- [62] M.A.S. Masoum, E.F. Fuchs, M.A.S. Masoum, E.F. Fuchs, Chapter 10 optimal placement and sizing of shunt capacitor banks in the presence of harmonics, in: power qual. Power syst, Electr. Mach. (2015) 887–959.

- [63] A. Águila Téllez, G. López, I. Isaac, J.W. González, Optimal reactive power compensation in electrical distribution systems with distributed resources. Review, Heliyon. 4 (2018).
- [64] R. Orizondo, R. Alves, UPFC simulation and control using the ATP/EMTP and MATLAB/simulink programs, in: 2006 IEEE PES Transm. Distrib. Conf. Expo. Lat. Am. TDC'06, Caracas, 2006.
- [65] I. Colak, S. Sagiroglu, G. Fulli, M. Yesilbudak, C.F. Covrig, A survey on the critical issues in smart grid technologies, Renew. Sustain. Energy Rev. 54 (2016) 396–405.
- [66] H. Bevrani, B. Francois, T. Ise, Microgrid dynamics and control, 2017.
- [67] L. Fan, Control and Dynamics in Power Systems and Microgrids, CRC Press, 2017.[68] J.A.P. Lopes, C.L. Moreira, A.G. Madureira, Defining control strategies for
- microgrids islanded operation, IEEE Trans. Power Syst. 21 (2006) 916–924.
  [69] N. Jaalam, N.A. Rahim, A.H.A. Bakar, C.K. Tan, A.M.A. Haidar, A comprehensive review of synchronization methods for grid-connected converters of renewable energy source, Renew. Sustain. Energy Rev. 59 (2016) 1471–1481.
- [70] E. Rokrok, M. Shafie-khah, J.P.S. Catalão, Review of primary voltage and frequency control methods for inverter-based islanded microgrids with distributed generation, Renew. Sustain. Energy Rev. 82 (2018) 3225–3235.
- [71] L. Priyadarshanee, Modeling and Control of Hybrid AC/DC Micro Grid, 2012, pp. 1–60.
- [72] F.D. Mohammadi, H. Keshtkar, A. Feliachi, State space modeling, analysis and distributed secondary frequency control of isolated microgrids, IEEE Trans. Energy Convers. 8969 (2017) 1–10.
- [73] M.A.A.M.S. Mahmouda, S. Azher Hussaina, Modeling and Control of Microgrid: an Overview, 2014.
- [74] F.H. Gandoman, A. Ahmadi, A.M. Sharaf, P. Siano, J. Pou, B. Hredzak, V.G. Agelidis, Review of FACTS technologies and applications for power quality in smart grids with renewable energy systems, Renew. Sustain. Energy Rev. 82 (2018) 502–514.
- [75] S.M. Dawoud, X. Lin, M.I. Okba, Hybrid renewable microgrid optimization techniques: a review, Renew. Sustain. Energy Rev. 82 (2018) 2039–2052.
- [76] R. Ayop, N.M. Isa, C.W. Tan, Components sizing of photovoltaic stand-alone system based on loss of power supply probability, Renew. Sustain. Energy Rev. 81 (2018) 2731–2743.
- [77] F.A. Khan, N. Pal, S.H. Saeed, Review of solar photovoltaic and wind hybrid energy systems for sizing strategies optimization techniques and cost analysis methodologies, Renew. Sustain. Energy Rev. 92 (2018) 937–947.
- [78] K.S. Rajesh, S.S. Dash, R. Rajagopal, R. Sridhar, A review on control of ac microgrid, Renew. Sustain. Energy Rev. 71 (2017) 814–819.
- [79] A. Bidram, S. Member, A. Davoudi, Hierarchical structure of microgrids control system, IEEE Trans. Smart Grid. 3 (2012) 1963–1976.
- [80] S. Sen, V. Kumar, Microgrid control: a comprehensive survey, Annu. Rev. Contr. 45 (2018) 118–151.
- [81] O. Palizban, K. Kauhaniemi, J.M. Guerrero, Microgrids in active network management - Part I: hierarchical control, energy storage, virtual power plants, and market participation, Renew. Sustain. Energy Rev. 36 (2014) 428–439.
- [82] O. Palizban, K. Kauhaniemi, Hierarchical control structure in microgrids with distributed generation : island and grid-connected mode, Renew. Sustain. Energy Rev. 44 (2015) 797–813.
- [83] A. Mehrizi-Sani, Distributed control techniques in microgrids, in: Microgrid Adv. Control Methods Renew. Energy Syst. Integr., 2016, pp. 43–62.
- [84] M. Saad, B. Arif, M.A. Hasan, Microgrid Architecture, Control, and Operation, Elsevier Ltd, 2018.
- [85] J. Hare, X. Shi, S. Gupta, A. Bazzi, Fault diagnostics in smart micro-grids: a survey, Renew. Sustain. Energy Rev. 60 (2016) 1114–1124.
- [86] S. Gholami, S. Saha, M. Aldeen, A cyber attack resilient control for distributed energy resources, in: Innov. Smart Grid Technol. Conf. Eur. (ISGT-Europe), 2017 IEEE PES, Torino, Italy, 2017.
- [87] V. Puig, J. Quevedo, T. Escobet, B. Morcego, C. Ocampo, Control Tolerante a Fallas (parte II): Mecanismo de Tolerancia y sistema supervisor, Rev. Iberoam. Automática e Informática Ind. 1 (2004) 15–31.
- [88] A. Mesbah, M. Kishida, R.D. Braatz, Design of multi-objective failure-tolerant control systems for infinite-dimensional systems, decis. Control (CDC), in: IEEE 52nd Annu. Conf., 2013, pp. 3006–3013.
- [89] A. Vargas-mart, L.E. Garza-casta, R. Morales-menendez, Sistemas de Control Tolerante a Fallas en Redes de Distribución de Energía, in: CLCA 2014, XVI Congreso Latinoamericano, 2014.
- [90] S. Saha, T.K. Roy, M.A. Mahmud, M.E. Haque, S.N. Islam, Sensor fault and cyber attack resilient operation of DC microgrids, Int. J. Electr. Power Energy Syst. 99 (2018) 540–554.
- [91] B. Li, Z. Shen, A Fault tolerant topology of inverter for micro- grid, in: Ind. Technol. (ICIT), 2016 IEEE Int. Conf., 2016.
- [92] V. Puig, J. Quevedo, T. Escobet, B. Morcego, C. Ocampo, Control Tolerante a Fallos (parte I): Fundamentos y Diagnóstico de Fallos, Rev. Iberoam. Automática e Informática Ind. 1 (2004) 15–31.
- [93] M.K.S. Khosravani, I.N. Moghaddam, A. Afshar, fault tolerant control of large power systems subject to sensor failure, in: 2015 Int. Conf. Power Eng. Energy Electr. Drives, 2011, pp. 1–7.
- [94] M.J. Morshed, A. Fekih, S. Member, A fault-tolerant control paradigm for microgrid-connected wind energy systems, IEEE Syst. J. (2016) 1–13.
- [95] H. Noura, D. Theilliol, J.-C. Ponsart, A. Chamseddine, Fault-tolerant Control System Desing and Practial Applications, 2009.
- [96] L. Wang, M. Cai, H. Zhang, F. Alsaadi, Active fault-tolerant control for wind turbine with simultaneous actuator and sensor faults, Complexity 2017 (2017) 11.

#### L. Ortiz et al.

- [97] M. Blanke, M. Kinnaert, J. Lunze, M. Staroswiecki, Diagnosis and Fault-Tolerant Control, Springer US, 2006.
- [98] M. Hosseinzadeh, F.R. Salmasi, fault-tolerant supervisory controller for a hybrid AC/DC micro-grid, IEEE Trans. Smart Grid (2016) 1–15.
- [99] L. Maharjan, T. Yamagishi, H. Akagi, J. Asakura, Fault-tolerant operation of a battery-energy-storage system based on a multilevel cascade PWM converter with star configuration, IEEE Trans. Power Electron. 25 (2010) 2386–2396.
- [100] S. Gholami, S. Saha, M. Aldeen, Fault tolerant control of electronically coupled distributed energy resources in microgrid systems, Int. J. Electr. Power Energy Syst. 95 (2018) 327–340.
- [101] S. Gholami, S. Saha, M. Aldeen, Sensor fault tolerant control of microgrid, in: IEEE (Ed.), Power Energy Soc. Gen. Meet. (PESGM), 2016, Power and Energy Society General Meeting (PESGM), 2016, 2016, pp. 1230–1235.
- [102] L. Yao, L. Li, C. Lei, Active Fault Tolerant Control for Nonlinear Non-gaussian Singular Stochastic Distribution Systems, 2016, pp. 450–455.
- [103] M. Morari, M. Mahmoud, J. Jiang, Y. Zhang, Active Fault Tolerant Control Systems: Stochastic Analysis and Synthesis, 2003.
- [104] A. Hooshyar, R. Iravani, Microgrid protection, Proc. IEEE 105 (2017) 1332–1353.
  [105] T. John, Y. Wang, T. Tan, et al., Model Predictive Control of Distributed
- Generation inverter in a microgrid, 2014 IEEE Innovative Smart Grid Technologies - Asia, ISGT ASIA 2014 (2014) 657–662.
- [106] S. Liu, W. Zhu, B. Xing, A model predictive control scheme for parallel-connected inverter-based distributed generators in micro-grids, 2015 23rd Mediterranean

Conference on Control and Automation, MED 2015 - Conference Proceedings (2015) 454–458.

- [107] X. Shi, J. Zhu, L. Li, Y. Qu, Model predictive control of PWM AC/DC converters for Bi-directional power flow control in microgrids, 2015 Australasian Universities Power Engineering Conference: Challenges for Future Grids, AUPEC 2015 (2015).
- [108] O. Abdel-Rahim, H. Funato, H. Junnosuke, Droop method based on model predictive control for DC microgrid, 19th International Conference on Electrical Machines and Systems, ICEMS 2016 (2016) 1–6.
- [109] S. Luo, C. Hu, Y. Zhang, et al., Multi-agent systems using model predictive control for coordinative optimization control of microgrid, 2017 20th International Conference on Electrical Machines and Systems, ICEMS 2017 (2017).
- [110] G. Lou, W. Gu, J. Wang, et al., Optimal Design for Distributed Secondary Voltage Control in Islanded Microgrids: Communication Topology and Controller, IEEE Transactions on Power Systems (2018).
- [111] X. Lu, X. Yu, J. Lai, et al., A Novel Distributed Secondary Coordination Control Approach for Islanded Microgrids, IEEE Transactions on Smart Grid (2016).
- [112] M. Savaghebi, A. Jalilian, J.C. Vasquez, J.M. Guerrero, Secondary Control Scheme for Voltage Unbalance Compensation in an Islanded, IEEE Transactions on Smart Grid (2012) 1–12.
- [113] S.K. Sahoo, A.K. Sinha, N.K. Kishore, Control Techniques in AC, DC, and Hybrid AC-DC Microgrid: A Review, IEEE Journal of Emerging and Selected Topics in Power Electronics (2017) 1–22.